

# ST. MARY'S UNIVERSITY COLLEGE SCHOOL OF GRADUATE STUDIES

# EMPLOYEES' PERFORMANCE EVALUATION PRACTICES AND CHALLENGES AT ZEMEN BANK

BY MULUGETA HAILU

> MARCH 2013 ADDIS ABABA, ETHIOPIA

# EMPLOYEES' PERFORMANCE EVALUATION PRACTICES AND CHALLENGES AT ZEMEN BANK

# BY MULUGETA HAILU

A THESIS SUBMITTED TO ST.MARY'S UNIVERSITY COLLEGE, SCHOOL OF GRADUATE STUDIES IN PARTIAL FULFILLMENT OF THE REQUIREMENTS FOR THE DEGREE OF MASTER OF BUSINESS ADMINISTRATION

MARCH, 2013 ADDIS ABABA, ETHIOPIA

# ST. MARY'S UNIVERSITY COLLEGE SCHOOL OF GRADUATE STUDIES FACULTY OF BUSINESS

# EMPLOYEES' PERFORMANCE EVALUATION PRACTICES AND CHALLENGES AT ZEMEN BANK

# BY MULUGETA HAILU

# APPROVED BY BOARD OF EXAMINERS

Dean, Graduate Studies	Signature
Advisor	Signature
External Examiner	Signature
Internal Examiner	Signature

# **TABLE OF CONTENTS**

D	eclaration	I
Er	ndorsement	II
A	scknowledgement	III
Α	bstract	IV
L	ists of Acronyms	V
L	ists of Tables	VI
С	HAPTER ONE	
1.	INTRODUCTION	2
	1.1. Background of the Study	2
	1.2. Statements of the Problem	4
	1.3. Research Questions	4
	1.4. Objective of the Study	5
	1.5. Significance of the Study/Problem	5
	1.6. Research Design and Methodology	6
	1.6.1.Research Design	6
	1.6.2.Population and Sampling Techniques	6
	1.6.3.Sources of Data Collection	7
	1.6.4.Data Analysis Method	7
	1.7. Delimitation/Scope of the Study	8
	1.8. Limitation of the Study	8
	1.9. Organization of the Study	8
C	HAPTER TWO	
2.	LITERATURE REVIEW	9
	2.1. Introduction	9
	2.2. Performance Appraisal an Overview	9
	2.3. Necessity of Performance Appraisal	10
	2.4 Methods of Performance Appraisal	11

	2.4.1.Traditio	nal Methods of Performance Appraisal Techniques	12
	2.4.1.1.	Ranking Straight	12
	2.4.1.2.	Comparing by Way of Paired Ranking	12
	2.4.1.3.	Assessment of Linear Rating Scale	13
	2.4.1.4.	Evaluation on the Parameters of Critical incidents	13
	2.4.1.5.	Evaluation by Group	14
	2.4.1.6.	Appraisal by Field Review	14
	2.4.2.Modern	Methods of Performance appraisal Techniques	14
	2.4.2.1.	Multi-level Multi-source Feedback System (360 Degree Feedback)	14
	2.4.2.2.	Evaluation on the Basis of Result Management by Objective (MBO)	15
	2.4.2.3.	Evaluation on Behaviorally Anchored Rating Scales	16
2.5	. Setting Perf	ormance Criteria	16
2.6	. Performanc	e Appraisal Processes / Guidelines	17
	2.6.1.Employ	ee Participation	17
	2.6.2.Develop	oing Performance Standards	17
	2.6.3.Goal Se	tting	17
	2.6.4.Sound F	Performance Appraisal Interview	17
	2.6.5.Self Eva	lluation	18
	2.6.6.Manage	ement Feedback	18
	2.6.7.Develop	o User-friendly Procedure	18
	2.6.8.Evaluat	or Training	18
	2.6.9.Revising	g Performance Appraisal Process	19
2.7	. Who Should	I Do the Appraisal	19
	2.7.1.Manage	ers/Supervisor Appraisal	19
	2.7.2.Self-App	praisal	20
	2.7.3.Subordi	nates Appraisal	20
	2.7.4.Peer Ap	praisal	20
	2.7.5.2Custon	ner Appraisal	21
2.8	. Benefits of P	erformance Appraisal	21

	2.8.1.Benefits	s for Employees	21
	2.8.2.Benefits	s for Supervisors	21
	2.8.3.Benefits	for the Organization	22
	2.9. Problems of	Performance Appraisal and How to Avoid Them	22
	2.9.1.Most Se	erious Performance Appraisal Problems	22
	2.9.2.Process	Related Problems	23
	2.9.3.Instrum	ent (form) Problems	24
	2.9.4.Manage	er/execution Problems	24
	2.9.5.Employ	ee/subject Problems	25
	2.9.6.Timing l	ssues	26
	2.10. Effective F	Performance Appraisal	26
	2.10.1. Relev	ance	26
	2.10.2. Sensiti	vity	26
	2.10.3. Reliab	ility	27
	2.10.4. Praction	cality	27
	CHAPTER THREE		
3	B. DATA PRESENTA	TION, ANALYSIS AND INTERPRETATION	28
	3.1. Demographi	c Information of the Respondents	28
	3.2. Analysis of f	inding of the study	30
	3.2.1.The Em	ployee Opinion on Performance Evaluation Criteria	30
	3.2.2.Employ	ees Opinion on Performance Appraisal Process at Zemen Bank	32
	3.2.3.Employ	ees Opinion on Performance Evaluation Problem	35
	3.2.3.1.	Employees' opinion on the most serious performance appraisal	
		Problem	35
	3.2.3.2.	Employees Opinion on Process Related Problem	36
	3.2.3.3.	Respondent Opinion on Instrument/Form Problem	37
	3.2.3.4.	Employees' Opinion on Employees Related Problem	39
	3.2.3.5.	Employees Opinion on Managers/Execution Problem	40

	3.3. Respondents/Managers Opinion on the Core Reason to conducted PA	. 43
	3.4. Respondents/Managers Opinion on the Real Problem Performance Appraisal	. 43
	3.5. Respondents/Managers Suggestion about Performance Appraisal Practices	. 44
	CHAPTER FOUR	
4.	SUMMARY, CONCLUSIONS AND RECOMMENDATIONS	. 45
	4.1. Summary of major findings	. 45
	4.2. Conclusions	. 46
	4.3. Recommendations	. 48
	Reference	. 50
	Annex I	. 53
	Annex II	. 58

# **DECLARATION**

I, the undersigned, declare that this thesis is my original work, prepared under the guidance of Ato Goitom Abraham (Asst. Prof.). All sources of materials used for the thesis have been duly acknowledged. I further confirm that the thesis has not been submitted either in part or in full to any other higher learning institution for the purpose of earning any degree.

Name Signature

St. Mary's University College, Addis Ababa February, 2013

# **ENDORSEMENT**

St Mary's University College Addis Ababa	February 2013
Advisor	Signature
	ersity auvisor.
Studies for examination with my approval as a unive	ersity advisor
This thesis has been submitted to St. Mary's Univer	rsity College, School of Graduate

#### **ACKNOWLEDGMENTS**

First and for most, my heartily thanks go to Almighty Father, God for his graciously provision of knowledge, wisdom, inspiration and diligence required for the successful completion of this paper and for bringing my dreams into reality. I am also greatly indebted to my advisor Ato Goitom Abraham (Asst. Prof.) for his countless suggestions, assistance and invaluable advice.

I would like to extend thanks to the staffs of Zemen Bank, who were cooperative and helpful; especially, Ato Alen Worku, who supported me to succeed in my academic progress.

Lastly but not the least, my warmest appreciation and thanks goes to my beloved family.

Thank you,

Mulugeta Hailu

#### Abstract

This project is designed to assess the Employees' Performance Evaluation Practice and challenges in Zemen Bank. On the basis of data collected through questionnaires and interview; the researcher have tried to reveal some of the real problems of performance evaluation practice based on the opinion of the respondents in this organization. The questionnaire was distributed to 75 employees of the bank working in head office and three kiosks: in which 53 were fully completed and returned. The questionnaire was distributed to the employees on the basis of simple random sampling based on respondent willingness and cooperation of the respondents. The data collected were analyzed using Microsoft Excel application software. On the basis of the data obtained from the respondents, Existence of quota system, Seniority, Subjectivity, lack of clarity of performance evaluation criteria, forms inadequacy to reflect the performance of the employees' jobs and duties, and manager related problems are the most serious problem identified at Zemen bank. In Addition, the researcher has discovered that evaluators usually do not continually record or document the performance of employees over the evaluation period. In this regard, it was identified that evaluators evaluate the performance of employees on the basis of recent behaviors'. In order to conclude, the performance evaluation system of the bank is attacked with problems. Based on the findings of the study, the researcher has forwarded some recommendations so that to give them an insight as to the practice and its associated problems of performance evaluation in the organization.

# **LIST OF ACRONYMS**

HR – Human Resource

MBO – Management by Objective

PA – Performance Appraisal

# LIST OF TABLES

Table 1	Respondents according to age and sex category
Table 2	Respondents According to Work Experience and Educational Background
Table 3	Employees opinion on Performance Evaluation Criteria
Table 4	Employees opinion on Performance appraisal process / Guide Line
Table 5	Respondents opinion on the most serious Performance appraisal problem
Table 6	Respondents opinion on Process Related problem
Table 7	Respondent opinion on instrument/form problem
Table 8	Respondents opinion on Employees / Subject Related problem
Table 9	Respondents opinion on Manager/Execution Related problem

#### CHAPTER ONE

#### INTRODUCTION

# 1.1. Background of the Study

Managing performance has been a very important issue for a long period of time. It has gained more attention recently due to high competitive business environment. Financial sectors especially Banks need to give great attention to their employees' performance and evaluation since employees are the first one who represents the bank in front of those prestigious customers. In addition, employees are responsibility to create excellent image about the bank to those customers in order to make the bank to be a leader in such competitive business environment. Hence "People have become the primary source of competitive advantage" (Wellins, Bernthal, & Phelps, 2005).

According to Longenecker and Fink (1999), formal evaluations are required to justify a wide range of human resource decisions like pay raises, promotions, demotions, terminations, training needs, and etc. They also emphasize that performance evaluation is one of the top 10 vehicles for creating competitive advantage in the business environment. Moreover, they assure that performance measurement allows the organization to tell their employees something about their rate of growth, their competencies, and their potentials.

On the other hand, Somerick (1993) has proven that ineffective performance evaluation can be a cause for many problems, such as low morale, low productivity, a lessening of an employee's enthusiasm and support for the organization, high rate of skilled employee termination, high rate of customer turnover, high running costs and etc that leads the company to shrink early.

In today's competitive business world, it is understood that organizations can only compete with their rivals by innovating, and organizations can be innovative by managing their human resources well. The human resource system can become more effective by having a valid and accurate performance appraisal system used for rating employees (Armstrong, 2003; Bohlander & Snell, 2004).

Performance evaluation demands the immediate supervisors to understand the nature of the job and the sources of information, and the information needs to be collected in a systematic way, and it is provided as a feedback, and integrated into organization's performance management process for use in making compensation, job placement, and training decisions and assignments (Islam and Rasad, 2005).

As reveled by Steers and Black (1994), "performance appraisal is one of the most important and often one of the most mishandled aspects of management." It has also been said to be one of the most problematic components of human resource management and is viewed as either a futile bureaucratic exercise or, worse, a destructive influence on the employee-supervisor relationship (Coutts and Schneider, 2004).

The usefulness of performance evaluation as a managerial decision tool depends partly on whether or not the performance appraisal system is able to provide accurate data on employee performance and hence rating accuracy is a critical aspect of the appraisal process. A difficulty of getting accurate appraisals of employee job behavior is most often attributed to: faults in rating format used, deficiencies in appraisal content, rater resistance to judge others, and the implications of the specific purpose of appraisal for the rater and the ratee (Decotiis and Petit, 1978).

Therefore, the problems of performance evaluation arise when the results of the evaluation fail to reflect the actual performance of the employees, which in turn, leads to wrong administrative decisions that can highly affect the life of the employees.

Thus, this paper tried to assess the practices and challenges of employees' performance evaluation in Zemen Bank S C. Hence such study has never been conducted before and the bank is new, encompassed with skilled manpower, innovative and well technologically equipped, it is advantageous for the bank to see early the problem occurring in relation with employees' performance evaluation and helps to take remedial action to overcome the cause and consequences identified.

# 1.2. Statements of the problem

As mentioned in the forgoing discussion, proper performance evaluation can have a number of advantages including job satisfaction, compensation, increase motivation, clear understanding of what is expected and what needs to be done to meet expectations, opportunity to discuss aspirations and any guidance, support or training needed to fulfill these aspirations, improved working relationships with the superior, opportunity to overcome the weaknesses by way of counseling and guidance from the superior, increased sense of personal value as employees are involved in the appraisal process, time will be devoted for discussing quality of work without regard to money issues.

On the other side, as different Authors ascertain in the above discussion poor performance evaluation causes so many problems. To mention, poor performance evaluation results in managers' inconsistency, evaluators' subjectivity, job dissatisfaction/low morale of employees, high retention rate of employees, inconsistency of feedback, poor employee/managers' relationship, hinder professional development, diminishes chances for merit raises. These arise due to irrelevance of the criteria used to evaluate the performance of the employees, lack of skills and knowledge of the raters, the subjectivity, favoritism and bias of the raters, inability to provide feedback as to the results of the performance evaluation.

Since, the above mentioned problems are inherent in every organization where there is a formally designed performance evaluation system and there is murmuring about the existence of those mentioned problem at Zemen Bank, this study addressed the problem occurred at Zemen Bank Due to poor performance evaluation practice, and the consequences' arise as a result of those problems.

## 1.3. Research questions

Based on the data collected from both management group and permanent employees of Zemen Bank, whose were evaluated for the last three years, the study tried to analyze the existence of the above mentioned problems.

This research tried to answer the following basic research question:-

- 1. How performance evaluation is conducted at Zemen bank?
- 2. For what purpose is performance evaluation mostly conducted by the bank?
- 3. What are the major performance appraisal problems of the bank?
- 4. What are the major consequences of performance evaluation practices on the overall employees' morale and retention?
- 5. What method should the bank use to improve the performance evaluation system in general?

# 1.4. Objective of the Study

The general objective of this study is to assess the practices and challenges of employees' performance evaluations process at Zemen Bank. Specifically, the objectives of the study are to:-

- 1. See the performance evaluation practice of the bank.
- 2. Determine the purposes of performance evaluation conducted mostly.
- 3. Identify the major performance appraisal related problem of the bank.
- 4. See whether employees' morale has been decreased and employees are leaving the bank due to performance evaluation practice.
- 5. Recommend alternative measures that should be taken based on the finding of the study.

# 1.5. Significance of the study/problem

Since performance evaluation does have various advantages for both organization and employees. such as pay increases, improvement and training, transfers, compensations, counseling, promotion, employee recognition, termination, and moreover enhancing firms overall performance, the outcome of this study supports by identifying the problems of performance evaluation and give signal to the Human Resource Management of the bank to take remedial action to minimize the barriers of performance evaluation in prospecting employees for different reasons as mentioned above.

This study also serves as a piece of contribution to the current knowledge in the practice of performance evaluation in an enterprise working in Ethiopia and may help to bring behavioral change in the areas of performance evaluation both in the mind of the raters, ratees and those parties responsible in the design of the instruments of performance evaluation forms that are used to judge the performance of employees. In addition, the result of the study may initiate the academia to conduct further performance evaluation related research in the future.

# 1.6. Research Design and Methodology

#### 1.6.1. Research Design

Since this study is a case study; it is conducted based on the descriptive research technique to describe the practices and challenges of employees performance evaluation in Zemen Bank S C. Because it is a descriptive research technique, questionnaires and interview have been used to collect data from employees and management group of the organization. Pilot test had been conducted among prospective employees prior to conducting the actual research in order to assure the effectiveness of the techniques.

#### 1.6.2. Population and Sampling Techniques

Currently, Zemen bank has around 150 (One Hundred Fifty) permanent employees out of 250 (Two Hundred Fifty) total employees, who are located at Head Office around Kazanchis and in three kiosks at Mexico area, Sweden embassy compound and Ambassador Area. This research targeted permanent employees of the bank addressing both at head office and kiosks, Hence performance evaluation focus on permanent employees only.

Simple random sampling techniques has been used, since the respondents are homogeneous and do have equal contribution to the study. Thus, it is difficult to address all permanent employees of the bank due to various constraints; this research addressed 75 employees of the bank, which is 50% of the total permanent employees of the bank and interview is conducted to 5 management group of the bank.

#### 1.6.3. Sources of Data Collection

The data were collected from two sources. 1) Primary data 2) Secondary data; Hence primary data are the main sources to the study; questionnaire containing Four parts 1) demographic aspect part 2) "Yes" or "No" part 3) Likert Scale Part 4) Personal Respond and comment parts, had be developed and distributed to management members of the Bank as well as permanent employees of the Bank. Moreover open ended interview had been developed and randomly conducted to both management group and permanent employees of the bank. To this end 75 questionnaires were distributed to employees of the bank who have been evaluated for the last one year and above and working in the main branches of the Bank and kiosks. The sample was selected from the total population of 150 permanent employees of the Bank. Among these questionnaires, 53 were fully completed and returned with a return rate of 71% which is acceptable.

Moreover, Interview was conducted with evaluators (Raters), employees and the Human Resource manager of the Bank through open ended questions which is designed to elicit the practices and challenges of performance evaluation of the bank. Among them 3 out of 5 employees of the bank answered the interview question.

The research analyzed secondary data through the detailed review of related literature. To this end, books, articles, journals, magazines, bulletins, broachers, and the company's Human Resource manual has been assessed and presented in the literature review part.

#### 1.6.4. Data Analysis method

Since, this study is descriptive research techniques; the data gathered are analyzed using SPSS computer program system to find out the overall result. In doing so Tables and percentages descriptive methods has applied to summarize and compare the final result.

# 1.7. Delimitation/Scope of the Study

This study tried to address both employees and management group of Zemen bank, which has only one branch and three Kiosks around Addis Ababa using questionnaires and interview. The study has focused on limited number of respondents; hence it is difficult to address all employees. The information gathered are analyzed using SPSS computer programming software, since the research is descriptive.

## 1.8. Limitation of the study

In addition to shortage of time and Budget, there were various reasons that limited the study. Like, among the expected number of employees, some of them were not willing to fill the questionnaires. Moreover getting access on employees' personal evaluation result of the previous year's records and unavailability of previous studies at international level on related topic had limited the research.

# 1.9. Organization of the study

The study is organized into four chapters. Chapter one deals with introduction, chapter two deals with the review of the related literature, chapter three deals with data analysis and interpretation and presentation and finally chapter four contains summary, conclusions and recommendations.

#### **CHAPTER TWO**

#### LITERATURE REVIEW

#### 2.1. Introduction

The main purpose of this chapter is to give a clear theoretical and practical understanding about employees' performance appraisal in business firms especially banks. Particularly, this paper focuses on ten major areas. The study starts with giving highlight on performance appraisal; based on the definition given by different scholarly. Then, necessity of performance appraisal; methods of performance appraisal; performance criteria's; benefits of performance appraisal; performance appraisal processes; effective performance appraisal are areas covered under this chapter.

## 2.2. Performance Appraisal an Overview

According to Lansbury (1988), performance appraisal is defined as "the process of identifying, evaluating and developing the work performance of the employee in the organization, so that organizational goals and objectives are effectively achieved while, at the same time, benefiting employees in terms of recognition, receiving feedback, and offering career guidance".

On the other hand, Baron and Armstrong (1998,38) define performance management as a systematic management tool comprised of purposes, standards, objectives, measurement, feedback and reward that are agreed on to receive more effective results from the organizations, teams and individuals by motivating the individuals in a way to make them be aware of their own potentials.

Moreover performance appraisal (PA) may be defined as a structured formal interaction between a subordinate and a supervisor where the subordinate's work performance is evaluated. It is a generic term for a variety of processes whereby an individual's work performance is assessed, usually by the person's closest supervisor, and discussed with a view to solving problems, improving performance and developing the individual appraised. Performance appraisals take place at

regular intervals and are usually based on a standardized method (agreed performance criteria and or measures). In the appraisal process, either the employee's performance or the outcome of performance is evaluated (Fletcher, 2003, 116; Dreves, 2002, 138; Lawler, 1990, 86).

## 2.3. Necessity of Performance Appraisal

Dewakar (2009, 20) has clearly explained the objective of performance appraisal in an organization. According to him the objectives of performance appraisal are categorized in to two major perspectives, namely, employees' perspectives and organization perspective. According to him, employees perceive positive benefit from the system when it is direct towards the consideration of an employee's needs while Organizations are demanding the system with the expectation of operational and business priority of the organization.

In addition, Dewakar has generalized that good performance appraisal will help to allocate resources, to reward for performance, to create a positive feedback, bring fairness, to identify training and development and create equal opportunity to all.

Similarly, Murphy (1995), states that the purpose of performance appraisal can be categorized basically in to two major perspectives, including organization's perspective and employees' perspective. He clarifies, organization's perspective as, to improve productivity and efficiency of an organization, to provide developmental feedback to employees, to document performance levels and decisions made, to achieve information on which to base other administrative decisions, to find the best fit between employees' qualifications and organizational tasks, to distinguish the best performers from poorer performers, to motivate poor performers, to maintain the motivation of good performers, to use appraisals for performance-based pay distribution and to articulate organizational goals to the individual level. Employees' perspective can be to improve performance or qualifications, to enable developmental feedback to the upper management, to promote career development and to receive rewards based on good performance.

Rynes (2005) also asserts that Performance appraisals are usually implemented for at least for two reasons. Firstly, performance appraisals are used as a management

tool in order to develop personnel. In particular, performance evaluation is seen to improve the performance of employees through developmental feedback (i.e. mainly focused on improving performance). Secondly, appraisals are used to make administrative decisions based on the evaluation of performance, namely, allocation of rewards and punishments such as pay increases, promotions or discharge.

We can generalize from the above discussion that formal evaluations of employee work behavior helps the employer and employee build on the strengths of the employee and identify those areas the employee needs improvement to be more effective and efficient in his/her job. Moreover performance evaluations can ensure the creation of reasonable performance standards so that both supervisor and employee are aware of work that is considered "acceptable performance." Because of the active involvement of both the supervisor and the employee in performance evaluations, an important channel of two-way communication will be also opened. Obviously, communication can result in increased cooperation and understanding between supervisors and employees, which in turn will enhance work performance and work environment thus providing better customer service to the community and each other.

# 2.4. Methods of Performance Appraisal

There are different performance appraisal methods thus, managers can choose from these appraisal methods while appraising the performance of their subordinates. This section deals with nine commonly used performance appraisal methods categorized under traditional methods and modern methods. These appraisal methods are briefly discussed below as asserted by different authors.

Traditional methods of performance appraisal techniques includes, ranking straight, comparing by way of paired ranking, assessment of linear rating scale, evaluation on the parameter of critical incidents, evaluation by the group and appraisal by field review.

Modern methods of performance appraisal techniques includes, multi-level multisource feedback system (360 degree feedback), evaluation on the basis of result or management by objective (MBO) and evaluation on behaviorally anchored rating scales.

#### 2.4.1. Traditional Methods of Performance appraisal techniques

As asserted by different authors some of the traditional performance appraisal techniques are explained below.

#### 2.4.1.1. Ranking Straight

This is one of the oldest and simplest techniques of performance appraisal. In this method, the appraiser ranks the employees from the best to the poorest on the basis of their overall performance and then the performance of one is compared with the other in all parameter. This method does have lot of drawbacks since peoples are different in behavior, way of working, style and concept of perfection, straight ranking techniques do not consider these factors (Dewakar 2009, 26-27).

According to Saiyadain (2004, 211), if an employee is evaluated by more than on supervisor, each one of the supervisors will rank the employee according to their own assessment and the mean will determine the employee performance. Comparison among the employees will be done by the mean result.

#### 2.4.1.2. Comparing by Way of Paired Ranking

As compared with ranking straight method this method is relatively better technique of comparison. Hence, the performance of one is compared with all members of the group. This method was short-lived because it was found to be quite difficult to design and it was time consuming and difficult to implement in a large number of people (Dewakar2009, 28).

The ranking techniques are determined by the number of time that the employee is superior over their co-worker. However, since this comparison is made on an overall basis and not in terms of specific job behavior or output, it may be subject to legal challenges (Casico 2004, 342).

#### 2.4.1.3. Assessment of Linear Rating Scale

This method is also known as graphic method since the parameters are putted in a graphic forms and tick box will be put in front of each parameters to rate employees from excellent performer to poor performer. The drawback of this method is that rater subjectivity is possible, But still most organizations throughout the world are using it and it is the most popular techniques (Dewakar 2009, 29).

Graphic methods of performance appraisal techniques helps to measure the overall competencies and objectives level the employees. In relation with competencies; the employee's potential to identify and analysis problems and his/her ability to maintain harmonious and effective work relationship with co-workers and constituents will be assessed. Pre stated objective will be evaluated in the objective parts (Dessler 2005, 317).

#### 2.4.1.4. Evaluation on the Parameter of Critical Incidents

In this method of Performance appraisal, the evaluator rates the employee on the basis of critical events and how the employee behaved during those incidents. It includes both negative and positive points. The drawback of this method is that the supervisor has to note down the critical incidents and the employee behavior as and when they occur (Dewakar 2009, 32).

The objective of critical incident method of performance appraisal technique is improve the supervisors ability as an observer and helps to train supervisors in taking notes while critical incidents on outstanding example of success or failure happens (Saiyadain 2004, 213).

This method helps the supervisor to provide actual examples of good and poor performance that use to explain employees rating, also it insure that the appraisal process cover the year. But this method is not too useful for comparing employees or making benefit decision (Dessler 2005, 322).

#### 2.4.1.5. Evaluation by the Group

As the name indicates this is a group evaluation technique. Here, the group will determine job performance standards and evaluate with the actual performances. Since the evaluation process is taken under a group it is time consuming but it eliminate the scope of any bias and it is quite simple (Dewakar 2009, 33).

#### 2.4.1.6. Appraisal by Field Review

In this method, the expert evaluator, a senior member of the HR department or a training officer discusses and interviews the supervisors to evaluate and rate their respective subordinates. A major drawback of this method is that it is a very time consuming method. But this method helps to reduce the superiors' personal bias (Dewakar 2009, 34).

In general, from the above discussion we can see that in a traditional performance appraisal approach only the supervisor or management groups or external experts are involving; but different scholarly have thought that involvement of employees is a necessity for a good performance appraisal.

#### 2.4.2. Modern Methods of Performance Appraisal Techniques

As asserted by different authors some of the modern methods of performance appraisal techniques are explained below.

# 2.4.2.1. Multi-level Multi-source Feedback System (360 Degree Feedback)

360 degree feedback techniques can be defined as a systematic collection and feedback of performance data on an individual or group derived from a number of stakeholders (the boss, team members, peers, customers, subordinates) in their performance (Dewakar 2009, 38).

360 Degree feedback system does work first by developing questionnaires, these questionnaires could be in a mixed or grouped forms, the respondents who are the stakeholders stated above do have a choice of alternative in relation to a particular

question, in addition even they can skip the rating. Moreover the individual who is going to be evaluated do have a chance to evaluate himself. Then all the respond will be collected summarized and conduct to the individual (Dewakar 2009, 37-40).

# 2.4.2.2. Evaluation on the Basis of Result or Management by Objective (MBO)

Here both the boss and the subordinates will be together to set realistic objective. First the boss give chance to the subordinates to set objectives then the boss decide as to what extent the subordinates should and plans the goal for the subordinate as per his expectation. After this, both the boss and the subordinates will have a mutual discussion about the objective to come in to a common agreement which will be written and signed by both (Dewakar 2009, 43).

The subordinate will be evaluated based on the objective agreed. This system does have a lot of advantage beyond other methods. Some of the advantages are, it motivates the subordinates, it integrates the entire organization since all are setting their objective towards the organization goal, and it helps employees to achieve their individual objective. However no system can be without its negatives. The drawback of this is the time involved in the process; it lays on sole emphasis on result rather than on quality traits in the performer (Dewakar (2009, 43-48).

According to Dessler (2005, 325) In order to implement management by objective appraisal techniques, there are six major steps;

Setting the organizational goal  $\rightarrow$  setting department goal  $\rightarrow$  discussing department goal  $\rightarrow$  define expected result  $\rightarrow$  performance review  $\rightarrow$  provide feedback

Management by objective performance appraisal techniques does have main positive impacts in the organization. Some of them are; it strength coordination among various unit in the organization; it becomes easy to implement hence the performer involved in setting up the goal; each employees are aware of what is expected from him/her; etc (Saiyadain 2004, 214).

In the other hand, according to Dessler (2005, 326), Management by objective performance appraisal techniques faces three major problems. 1) Setting unclear

and non measurable objective 2) It is highly time consuming 3) Setting objective with the subordinate sometimes turn in to a tug-of-war.

#### 2.4.2.3. Evaluation on Behaviorally Anchored Rating Scales

This technique is based on the concept that one can assess an employee's performance by evaluating his behavior or action over a particular situation. The parameters used in this technique are human relation, consciousness, organizational ability, observational power, knowledge and judgments. This technique is quite time consuming and expensive. Moreover it has no clear cut advantage as compare with the other, but it is quite modern to compare the competence of an individual (Dewakar 2009, 49).

Behaviorally anchored rating scales performance appraisal techniques basically combine the benefit of critical incident methods and graphic rating scale methods by anchoring a rating scale with specific behavioral example of good or poor performance. This technique provide better, more equiTable appraisal than the others (Dessler 2005, 322).

According to Dessler (2005, 323-324), behaviorally anchored rating scales techniques is implemented through five basic steps;

Generate critical incidents  $\rightarrow$  developing performance dimension  $\rightarrow$  reallocate incidents  $\rightarrow$  scale the incidents  $\rightarrow$  develop a final instruments

Even though this technique is time consuming, it has some advantages than others. Result in more accurate gauges, clearer standards, feedback, independent dimension and consistency are some of the advantage that makes the system different from the others (Dessler 2005, 324).

# 2.5. Setting Performance Criteria

As pointed out by Vassallo (2004, 277), the criteria (dimensions) selected in forming an evaluation system should meet seven qualifications. Some of the factors observed in this study are, *validity*: criteria should be appropriate for the project and aligned with the objectives and goals, *directness*: criteria should be stated as

clearly as possible, *objectivity*: criteria should be measurable, *adequacy*: criteria should sufficiently measure the outcome, *quantitativeness*: criteria should be quantitative whenever possible, *practicality*: criteria should be obtainable efficiently and at a reasonable cost and *reliability*: criteria should be designed to enable replicable results.

#### 2.6. Performance Appraisal Processes / Guidelines

According to Islam and Rasad (2005), employees' performance management techniques should have to consider the following areas in order to come up with good performance appraisal techniques.

#### 2.6.1. Employee Participation

In any case, if the employees perceive the appraisal system as biased, unfair and lacks rigors, then it is unlikely that they will accept the outcomes of the system. Participation gives an opportunity to the employees to raise their voice into the appraisal process. Performance standards, criteria for evaluation and the evaluation form itself – all can be developed with the help of employees (Islam et al 2005).

#### 2.6.2. Developing Performance Standards

When developing standards, they must be essential to measure the job duties and responsibilities. In addition, participation of employees will make the standard reliable, valid, fair and useful performance standards (Islam et al 2005).

#### 2.6.3. Goal Setting

Setting goal specifically performance measurement goal will motivate employees and create appraisal satisfaction, especially when the criteria's are specific, measurable, moderately challenging and acceptable (Islam et al 2005).

#### 2.6.4. Sound Performance Appraisal Interview

The appraisal interview must be conducted properly in order to get the most out it. The interviewer must be aware about sensitivity to employee needs for privacy and confidentiality. It is of utmost importance to provide undivided attention during the interview and reserve adequate time for a full discussion of the issues. Moreover an open ended question increase participation of employees in the evaluation processes (Islam et al 2005).

#### 2.6.5. Self Evaluation

Self evaluation provides employees an opportunity to systematically assess their performance. Studies have indicated that self evaluation increases employees' perceived fairness on the appraisal process. Employees can evaluate themselves by completing their own appraisal form and presenting the draft for discussion with the evaluator (Islam et al 2005).

#### 2.6.6. Management Feedback

Management's feedback is required for a common sense reason. When the employees do good jobs, they expect a pat on their backs (positive feedback); on the other hand, if the poor performers do not receive any constructive feedback which tells them to improve, they will think that the present level of performance is accepted in the organization and they may not put extra efforts to improve (Islam et al 2005).

#### 2.6.7. Develop User-friendly Procedure

Performance criteria and rating procedures should be simple enough and they should be well understood by the raters and ratees. Performance criteria should encompass the key aspects of employee's job. If any key aspect is ignored, then it sends the message that is unimportant and can be ignored. Appraisal system will be successful only when the items appraised address the requirements and essential functions on the job. The criteria used should be specific and directly related to the job (Islam et al 2005).

#### 2.6.8. Evaluator Training

The person who conducts the appraisal exercise should receive extensive training in goal setting, setting performance standards, conducting interviews, providing

feedback, avoiding rating biases, etc. He/she should know how to conduct appraisals accurately, consistently, fairly and objectively. In particular, top management must be aware about the competency level of the raters (Islam et al 2005).

#### 2.6.9. Revising Performance Appraisal Process

The performance appraisal result should be revised by the responsible department in order to assure that the process and practices are in line with the rule and regulation of the organization. By revising the process any bias or subjectivity can be avoided and a good performance appraisal technique will be implemented (Islam et al 2005).

From the above discussion it is clear that good performance appraisal activities should give attention to employees participation, setting pre determined performance standards, designing well defined goal, the arrangement of sound performance appraisal interview, the implementation of self evaluation, the need of providing management feedback, development of user-friendly procedure, evaluator training and revise performance appraisal process.

# 2.7. Who Should Do the Appraisal

Previously most organization were doing performance appraisal by a single person. But now a day given the complexity of today's jobs, it is often unrealistic to presume that one person can fully observe and evaluate an employee's performance. The raters may include supervisors, peers, team members, self, subordinates, and customers (George B & Scott S 2009; Cascio 2004, 348).

# 2.7.1. Managers/Supervisor Appraisal

This method of appraisal technique is traditional. Usually supervisors evaluate the performance of an individuals and then supervisor superior review the evaluation and appraise it. The negative impact of this technique is that when the supervisors face shortage of time to observe the performance of the individual; they simply relay on the past performance records of the individual. Even if reviews by superiors generally are more objective and provide a broader perspective of employee

performance it is highly subjective to the supervisor (George et al, 2009; Cascio 2004, 348).

#### 2.7.2. Self-Appraisal

This appraisal methods increase the involvements of the individual in the appraisal process. Hence the employee is given a chance to complete the appraisal form prior to the review process it makes the job performance discussion among them simple and helps to agree on final appraisal. This approach also works well when the manager and the employee jointly establish future performance goals or employee development plans (George et al, 2009; Cascio 2004, 348).

#### 2.7.3. Subordinates Appraisal

This is appraisal techniques made by subordinates to their managers. Most of the time managers are hesitant to endorse such a system, particularly when it might be use as a basis for compensation decisions. However, when the information is used for developmental purposes, managers tend to be more open to the idea. These techniques will be advantageous to manager to improve their performance if they heed the advice of their subordinates (George et al, 2009; Cascio 2004, 348).

#### 2.7.4. Peer Appraisal

Peers appraisal technique is evaluation of performance of one another who are horizontal in the organization structure. This will help to see different dimension of performance hence peers can readily identify leadership and interpersonal skills along with other strengths and weakness of their co-workers. However, peer appraisals may not be advisable for administrative decisions such as salary or bonuses. Employers using peer appraisals must also be sure to safeguard confidentiality in handling the review forms. Any breach of confidentiality can create interpersonal rivalries or hurt feelings and bring about hostility among fellow employees (George et al, 2009; Cascio 2004, 348).

#### 2.7.5. Customer Appraisal

The Customer performance appraisal system is like team appraisal, which is based on TQM concepts and seeks evaluation from both external and internal customers. For both developmental and administrative purposes, internal customers can provide extremely useful feedback about the value added by an employee or team of employees (George et al, 2009; Cascio 2004, 348).

# 2.8. Benefits of Performance Appraisal

As asserted by Charlie (2011), performance appraisal programs are important for the employees, supervisors who conduct appraisals, as well as companies. Brief explanation of these benefits is below.

#### 2.8.1. Benefits for Employees

Employees are beneficiary in performance management program with many dimensions. Some of these are; It help to create clear understanding of his/her role and what is expected from her/him in the organization, helps to know his/her strength and weakness and to improve them, it is a means to get opportunities for further career development. Moreover, they get to know the way the performance appraisal process is conducted by their supervisors. Therefore, in order to take the advantage individual will focus on his work and try to deliver quality works (Charlie, 2011).

On the other hand Saiyadain (2004, 203) has asserts that employees are beneficiaries of performance appraisal through personal development, satisfaction and involvement of the individuals, and the perception of fair and just compensation.

#### 2.8.2. Benefits for Supervisors

As employees are beneficiary in the performance management program; supervisors also do have many benefit packages. For instance; supervisors will get time to interact with the employees and understand their difficulties, helps to judge employees' confidence level and spot their difficulties, it helps to identify level of human resource and the gap and the supervisor gets to know of the training and development needs of an employee (Charlie, 2011).

#### 2.8.3. Benefits for the Organization

Performance appraisal program do have great advantage to the organization. Through performance appraisal processes organization will get the chance to identify candidates with high potentials, helps to know the expectation of thee employees from the organization, help organizations to improve the overall workforce efficiency, skills and productivity and to build good relationships with each employee, help to decide how successful the company's induction and recruitment policies are, help to achieve organizational plan and etc (Charlie, 2011).

Moreover Saiyadain (2004, 203) has explained that organizations are benefited with performance appraisal through generating man power information, and provide a rational compensation structure.

# 2.9. Problems of Performance Appraisal and How to Avoid Them

John Sullivan with his magazine has explained the top 50 problems with performance appraisal. He categorized them as, most serious performance appraisal problems, process related problems, instrument (form) problems, manager/execution problems, employee/subject problems and timing issues. With his explanation he has clearly shown us that evaluating employee performance has many dimensions of difficulties which need to give serious attention. (http://www.tlnt.com) Some of these performance appraisal problems those related with this study are briefly discussed in the following section;

#### 2.9.1. Most Serious Performance Appraisal Problems

According to the Author mentioned above some of the most serious performance appraisal problems include the following.

**Don't assess actual performance: -** Most of the time managers assess to factors that contributes to actual performance like characterizations of their personal "traits" (i.e. commitment), knowledge (i.e. technical knowledge) or behaviors (i.e. attendance) rather than assessing the actual output.

**Infrequent feedback:** - feedback is given at the time of conducting performance result, but feedback need to be given at formal interval before conducting performance appraisal result.

**Non-data-based assessment:** - most performance appraisal processes are subjective due to not having pre populating forms with data to evaluate individuals. Decision made based on such subjectivity leads to poor performance appraisal process and creates bias among the individuals.

Lack of accountability: - most of the time managers are not accountable for their in accurate feedback or for making mistakes. If an individual keep silent with the wrong feedback given by his immediate supervisor no one will ask the supervisor his/her reason for the wrong feedback. In practice the supervisor is asked when the question is come from the individual.

#### 2.9.2. Process Related Problems

As per Sullivan magazine process related problems includes the following.

**Disconnected from rewards:** - sometimes performance appraisal is not liked with employees benefit package or punishment like getting a merit raise, bonus, or promotion, demotion etc. At this time supervisors and manager may take the performance appraisal process carelessly.

**Each year stands alone:** - each performance appraisal by definition covers a finite period of time. However, if the goal is to assess potential and identify patterns, an employee's performance must be assessed over multiple years.

A focus on the squeaky wheel: - most performance appraisal systems focus on weak performers. There is significantly less focus on top performers and thus there is no system to capture their best practices and then to share them with others.

**No second review: -** in most organization performance appraisal is conducted by a single supervisor and conducted to the individuals. Since performance appraisal is conducted for the benefit of both individual and organization; to get accurate and fairness result a second review is necessary.

**Cross-comparisons are not required: -** most of the time performance appraisal processes are not done by comparing each member of the team with one another. A cross comparison is require in order to identify the strongest and weakest side of one another.

#### 2.9.3. Instrument (form) Problems

As ascertained by Sullivan in his magazine, Instrument (form) related problems of performance evaluation process includes,

**Doesn't address diversity: -** all too often, the same appraisal form is applied to a large but not homogeneous group of employees (i.e. all hourly, all exempts, all managers etc.). As a result, the assessment form does not fit the job.

The factors are all equal: - most forms treat all assessment factors as if they are of equal importance. Instead, they should be weighted based on their relative importance in a particular job.

**Disconnected from job descriptions:** – in many cases, the factors on the form are completely different from the factors on an employee's job description, bonus criteria, or yearly goals. This can confuse employees and cause them to lose focus.

#### 2.9.4. Manager/execution Problems

Managers related problems are the most briefly stated problems of performance evaluation process by the Author. Below are some of common managers' related problems explained by the Authors mentioned above.

**Managers are not trained:** - in most organizations, managers are not trained on how to assess and give honest feedback. If the process includes a career development component, it is even more likely that managers will not know how to enhance the career path of their employees.

**Gaming the system: -** often managers artificially rate individual employees to save money or to keep employees from becoming visible for promotion. Some selfishly

give a score just below that required for a pay increase, while others give scores just above the point where they would be required to take disciplinary action.

**Recency errors:** - managers, especially those who don't consult employee files and data, have a tendency to evaluate based primarily on events that occurred during the last few months (rather than over the entire year).

**Inconsistency across managers:** - some managers are naturally "easy raters" while others are not. As a result, employees working under easy managers have a better chance of promotion due to their higher scores. In firms that rely heavily on the narrative portion of the assessment, having a manager with poor writing skills may hamper an employee's career. Without "benchmark" numbers to set as a standard, inconsistency is guaranteed in large organizations.

**Managers don't know the employee:** - managers of large and global organizations, as well as newly hired and "transferred in" managers may be forced to do appraisals on employees they barely know. Recently promoted managers may be forced to assess their former friends and colleagues.

# 2.9.5. Employee/subject Problems

Employees related problems are also briefly explained by the author In addition to the above mentioned types of PA problems. Some of the problems raised in relation with employees are:-

**One-way communication:** - some managers simply give the employee the form to quickly sign and they don't even solicit feedback. Many employees are intimidated by managers and the process, and as a result, they say nothing during or after the appraisal.

**Self-assessment is not possible:** - Generally employees do not have a chance to evaluate themselves. It is possible to say that this practice is not implemented yet. But giving a chance to self assessment will create a smooth environment and end up with good performance appraisal result.

**One-way process: -** in most cases, employees do not have information about the factors on how they will be evaluated. Managers conduct the evaluation process in on way. There is no formal and informal communication between employees and managers prior to conducting the appraisal.

**Retention issues:** - employees' retention issue rose when the performance appraisal process is conducted unfairly. Top employees are turns to look for another job when they feel that the performance appraisal is unfair and no differential in recognition and rewards for their superior performance.

**Many possible emotional consequences: -** if performance appraisal is blotched, you can expect a decrease in employee engagement, trust employer brand strength, teamwork, and innovation contribution. Employee referrals from disgruntled employees will probably also drop.

### 2.9.6. Timing Issues

A time-consuming process: - most of the forms are incredibly long and time-consuming. As a result, some managers routinely recycle "last year's" evaluations. If HR is required to sit in on the sessions, the amount of wasted time increases significantly.

# 2.10. Effective Performance Appraisal

As asserted by Saiyadain (2003, 336) effective performance appraisal systems that are applicable in this research are:-

### 2.10.1. Relevance

As asserted by the author, the performance standards/criteria should be clearly linked with the organization goal and the employee's specific job duties and responsibilities. Moreover the performance standards / criteria should measure how well an employee has done the work. He has defined relevance as

"What really makes the difference between success and failure on a particular job, and according to whom?" Saiyadain (2003, 336)

From the above definition we can understand that the performance evaluation criteria in addition to linked with the goal and objective of the organization, it should have to measure the performance /achievement of employees out com.

### 2.10.2. Sensitivity

According to Saiyadain (2003, 336), the performance appraisal system should have to differentiate between good performer and poor performer. While doing so, employees' morale will be built and the performance appraisal result can be used for administrative purposes. If not; the process will be pointless work.

### 2.10.3. Reliability

This is the most important requirement of effective performance appraisal techniques. The performance evaluation result done by different parts (such as, Supervisors, Peers, Subordinates) should have to agree closely with one another. In order to do this all the evaluator need to have enough time to observe the individual. Saiyadain (2003, 337)

### 2.10.4. Practicality

The performance appraisal process of an organization need to be as simple as to put in to practice, Moreover the outcome of the result needs to serve its purpose. The purpose could be employees' promotion, bonus, salary increment and employee's recognition or demotion. (Saiyadain 2003, 337)

### **CHAPTER THREE**

### DATA PRESENTATION, ANALYSIS AND INTERPRETATION

This chapter presents the analysis and interpretation of the data collected from both primary and secondary sources. In the first part discusses the demographic information of the respondents and the second part discusses the information obtained from the respondents and the secondary data source consulted.

# 3.1. Demographic Information of the Respondents

The first part of the questionnaire consists of the demographic information of the participants. This part of the questionnaire requested a limited amount of information related to personal characteristics of respondents. Accordingly, the following variables about the respondents were summarized and described in the subsequent Tables. These variables include age, sex, work experience and educational background.

Table 1 Respondents according to age and sex category

			Sex		Total			
Age		Male	F	emale		Otal		
	No.	%age	No.	%age	No.	%age		
Under 25	6	11%	4	8%	10	19%		
25-34	15	28%	9	17%	24	45%		
35-44	7	13%	4	8%	11	21%		
45-54	2	3.75%	2	3.75%	4	7.5%		
>55	2	3.75%	2	3.75%	4	7.5%		
Total	32	60%	21	40%	53	100%		

As indicated in Table 3.1.above, about 60% of the respondents were male and the remaining 40% are female. From the total respondents, majority of them found in the age of range 25-34 years. This age group covers 45 % of the total respondents. On the other hand, 21% of the respondents belong to the age range of 35-44 years.

Quite considerable numbers of respondents, i.e., 19% were also under the age of 25, those who were above the age of 45 years counts to 16%.

As can be realized from the data, most of the respondents are young. Thus majority of the bank's employee are in the productive age, helps the bank to design a sound working system, since they can be trained, educated and developed. Easily by doing so, the bank will gain a competitive advantage in the banking industries

Table 2 Respondents According to Work Experience and Educational Background

Work Experience			Ed	ucational	l Qualif	ication		
	Dip	loma	First	Degree	Ма	sters	Т	otal
	No.	%age	No.	%age	No.	%age	No.	%age
1-2 Yr	1	2%	10	19%	0	0%	11	21%
2-3 Yr	4	7.5%	8	15%	0	0%	12	22.5%
3-4 Yr	2	4%	9	17%	0	0%	11	21%
4 years and above	4	7.5%	14	26%	1	2%	19	35.5%

As can be seen from Table 3.2 above most of the respondents (26%) are first degree holders and worked for more than 4 years at the bank, whereas 19% of the respondents are first degree holder and worked for 1-2 year. Of the respondents 17% are also first degree holder worked 3-4 years in the bank. Quite considerable number of respondents i.e., 7.5% are found diploma holder worked for 2-3 years. There is only one respondent of masters holder worked for 4 years that covers 2% of the total respondents.

In general, all respondents are professional and worked at the bank for 1 year and above; this makes the responses to be valid since most performance evaluation process is conducted to permanent and professional employees of the bank worked for more than 1 year and above.

## 3.2. Analysis of finding of the study

This section presents the analysis of data collection in detail:-

## 3.2.1. The Employee Opinion on Performance Evaluation Criteria

Performance evaluation criteria need to in line with the goal and objective of the organization, it must be measurable, should be clearly stated and directly related to employees job duties and responsibilities. Since designing performance evaluation criteria is the first and core part of the evaluation process, giving serious attention to this issue will determine the effectiveness of the performance evaluation practices of the bank. In practice, many organizations use the same criteria to evaluate huge number of employees at the same weight. Using the same form with equal weight makes the output of the performance evaluation result irrelevant and employees will be dissatisfied.

In this respect, the opinion of respondents about the criteria used at Zemen Bank has been summarized and described in Table 3 below.

Table 3 Employees opinion on Performance Evaluation Criteria

Items		Strongly Agree		Agree		Neutral		Disagree		Strongly Disagree		Total	
	No.	% age	No.	% age	No.	% age	No.	% age	No.	% age	No.	% age	
The Criteria's are in line with the bank goal	9	17%	27	51%	8	15%	59	9%	4	8%	53	100%	
All Employees are evaluated with the same forms	12	23%	14	26%	15	28%	7	13%	5	9%	53	100%	
All criteria's are equally weighted	5	9%	7	13%	3	6%	27	51%	11	21%	53	100%	

Based on the responses gathered from the employees of the bank, the researcher has tried to discuss the employees' attitude towards the performance evaluation criteria's. The questionnaires were designed using Likert Scale where almost all the statements were measured on a five point scale with 1 = strongly Agree; 2 = agree; 3 = neither agree nor disagree; 4 = disagree; and, 5 = strongly disagree. The information obtained from the questionnaire are summarized and discussed in this section.

As we can see from the above Table, 68% of the total respondents agreed that the criteria are in line with the bank goal; from this 17% are strongly agreed. 15 % of the total respondents are neutral to the statement. But 17% of the total respondents disagreed that the criteria are in line with the goal and objective of the bank; of which 8% of them strongly disagreed. From the responses of employees we can conclude that Zemen bank performance evaluation criteria are in line with the bank goal.

When we see the respondents' opinion regarding "All employees are evaluated with the same form/criteria", 23% of the total respondents strongly agreed and 26% agreed. 28% of the respondents are neutral. Whereas 9% strongly disagree with the statements and 13% disagree. This shows that Zemen bank performance evaluation form is same and there is no form differentiation between employees, in addition employees are evaluated with same forms. But in order to measure employee output, the criteria/form needs to be designed according to the employee's job duties and responsibilities.

According to the information gather from respondents, 9% strongly agreed that the criteria's are equally weighted whereas 13% agreed. From the respondents 6% are neutral to the item. But a total of 72% disagreed that the criteria's are equally weighted. This implies that performance evaluation criteria have given different weigh. Since the contribution of each criteria towards the performance of an employee is different, giving different value is a positive side of the bank.

In general, Zemen bank performance evaluation criteria are in line with the bank goal and objective and point given to each criteria is in accordance with the general rule of performance evaluation criteria. But, these criteria are not specific and directly

related with the employees' job and duties, they are not sufficient enough to measure the performance of an individual.

# 3.2.2. Employees Opinion on Performance Appraisal Process at Zemen Bank

In any organization, the performance evaluation process should include some basic guidelines. To mention, the evaluation processes should give employees awareness on how performance appraisal process is conducted, evaluators should assign time to interview employees to hear and address employees related issues, managers should give feedback during a job and at the time of conducting performance evaluation result, self evaluation should be practiced by the employee prior to conducting the final result and the final result need to be evaluate by independent party in order to assure faultlessness.

The following Table shows the summery of the total respondents about the performance appraisal process conducted at Zemen Bank.

Table 4 Employees opinion on Performance appraisal process / Guide Line

Items		ongly gree	Agree		Neutral		Disagree		Strongly Disagree		Total	
	No.	% age	No.	% age	No.	% age	No.	% age	No.	% age	No.	% age
Awareness of employees about PA process	3	6%	16	30%	15	28%	9	17%	10	19%	53	100%
Time given to interview employees'.	5	9%	15	28%	12	23%	9	17%	12	23%	53	100%
Self Evaluation is not Allowed	7	13%	8	15%	7	13%	16	30%	15	28%	53	100%
Feedback given during a job	4	8%	10	19%	15	28%	30	19%	8	15%	53	100%
Results are revised by independent part	5	9%	8	15%	15	28%	12	23%	13	25%	53	100%
Feedback Given at Conducting PA Result	8	15%	20	38%	11	21%	6	11%	8	15%	53	100%

As we can observe from the above Table, 30% of the total respondents agreed that employees are aware of how the performance appraisal process is conducted while 6% strongly agreed. 28% of the respondents are neutral to the issue. But a total of 36% of the respondents disagree the statement; from this 19% are strongly disagreed. Even if considerable numbers of respondents disagree with the statement "Zemen bank does not create awareness among employees towards how performance appraisal process is conducted", it is not at the required level.

According to information gathered from respondents, a total of 40% disagree that managers are giving time to hear and address employees with performance related issue, while 37% of respondents agree. 23% of the total respondents are neutral. This implies that managers are not giving time to hear and address employees' problem, since this is one of the techniques that help managers to address sensitive and confidential information about employees that hinder their performance, managers are required to give time and full attention to interview and hear their employees. Easily by doing so, employee's performance will increase as well as organization performance.

On the other hand, 28% of the total respondents agree that employees are not allowed to evaluate themselves, where as 30% agree the statements. Only 13% of the total respondents strongly agree that self evaluation is allowed in the bank. 13% are neutral. Self evaluations help employees to perceive fairness on the evaluation process. It creates a chance to discus evaluators with employee. As the above data shows, the level of self evaluation practice at Zemen bank is poor, human resource department need to design a technique that employees will evaluate themselves.

A total of 15% of the respondents agree that feedback is not given during a job, whereas 30% of the total respondents' agree. However, 28% of the respondents are neutral to the statement "feedback is given during a job". But 38% agree that managers give feedback during a job, of which 8% are strongly agreed. This shows that, on average managers are not giving adequate feedback to employees on their performance. Giving feedback help employees to know their strength and to improve their weakness, if managers keep silent from giving feedback, employees will think that their present level of performance is acceptable in the organization and they may not put extra efforts to improve.

Most of the respondents (25%) strongly disagree that the evaluation result is revised by responsible department, where 23% of the total respondents disagree with the statement. Only 9% of the total respondents strongly agree that the evaluation result is revised by responsible department and 15% agree the statements. 28% of the respondents are neutral to the item. In order to assure that the performance evaluation process is in accordance with the rule and regulation of the bank and to avoid and bias and subjectivity, independent department need to evaluate the performance evaluation result of employee prior to conducting to employees. The above summary shows that Zemen bank is at early stage to revise performance evaluation result by independent department.

As it is possible to observe from the above Table, 15% of the total respondents strongly agree that feedback is give at the time of conducting performance result only, 15% of the respondents also agree with this statement, whereas a total of 26% disagree the item. Moreover, 21% of the total respondents are neutral. From the data we can see that feedback is given only the time of conducting performance evaluation result. Even if giving feedback at the time of conducting appraisal is in accordance with the general rule, giving at that time only will affect employees as well as organizational performance since employee will not have a chance to improve their weakness at the right time.

In order to summarize, Zemen bank need to improve the performance evaluation processes. Since employees are not aware of how performance appraisal is conducted, managers do not give time to hear and interview employee about performance related issues, there is no independent department who revise the performance appraisal processes and one of the most required processes which are self assessment is not practiced yet. Moreover, the frequency of feedback given during a job is not significant. In addition feedback is mostly given only at the time of performance appraisal conducted.

# 3.2.3. Employees Opinion on Performance Evaluation Problem

# 3.2.3.1. Employees opinion on the most serious performance appraisal problem

Most serious performance appraisal problem mainly focus on giving feedback and accountability. In performance appraisal processes managers are expected to give positive feedback about employees' good performance and constructive feedback to improve poor performance of employees. Beside, managers need to be accountable for their wrong feedback and subjectivity.

The following Table shows us that to what extent these problems exist at Zemen Bank.

**Table 5** Respondents opinion on the most serious Performance appraisal problem

Items	Strongly Agree		Agree		Neutral		Disagree		Strongly Disagree		Total	
	No.	% age	No.	% age	No.	% age	No.	% age	No.	% age	No.	% age
Feedback given to employee'.	8	15%	6	11%	13	25%	5	9%	21	40%	53	100%
Managers are not accountable.	10	19%	14	26%	15	28%	10	19%	4	8%	53	100%

The above Table indicates that the majority of the respondents (49%) disagree with the statement "Feedback given to employee so that they know where they stand" and only 26% of the respondents' agree. 25% of the respondents are neutral with the statement. Giving feedback is one of the core techniques that helps o improve employees' performance and helps employees to know their strength and improve their weakness, Managers should give feedback at frequent interval. Moreover two way communication between employees and managers will be practiced. If managers are not at the required level to give feedback, employees will consider that their performance level is acceptable and enough to the achievement organization

goal. The above Table indicates that Zemen bank managers are not at the required level to giving feedback. In order to improve employees' performance managers need to increase frequency of giving feedback.

Since managers are not accountable for their wrong practice of performance evaluation process and feedback is not given to employees to strengthen employees' performance and to give flash to their weak performance, we can conclude that Zemen bank is attacked with most serious performance related problems.

# 3.2.3.2. Employees Opinion on Process Related Problem

In order to avoid process related problems, employees need to be evaluated by comparing with their co-workers, the evaluator need to focus on employees' whole year performance rather than a few day performances and employees performance appraisal process should be conducted based on multiple result.

The following Table shows summary of to what extent Zemen Bank is facing with process related problem of performance evaluation practice.

Table 6 Respondents opinion on Process Related problem

Items		Strongly Agree		Agree		Neutral		Disagree		ongly agree	Total	
	No.	% age	No.	% age	No.	% age	No.	% age	No.	% age	No.	% age
employees evaluation with co-worker	6	11%	17	32%	15	28%	11	21%	4	8%	53	100%
Evaluation focus on the whole year performance	9	17%	6	11%	12	23%	14	26%	12	23%	53	100%
Evaluation conducted based on multiple year result.	10	19%	5	9%	12	23%	6	11%	20	38%	53	100%

As we can observe from the above Table, trend of Zemen bank evaluating employees by comparing with their co-workers is good. Most employees (43%) agree the point, while insignificant amount of respondents strongly disagree. 28% of the total respondents are neutral to the statement. Employees' evaluation done by comparing with one another will helps to identify the strongest and weakest side of one another moreover it create competitive environment among employee.

On the other hand, 49% of the total respondent disagree that evaluation focus on the whole year performance where 28% agree. 23% of the total respondents are neutral to the statement. This implies that most evaluators focus on few day performances of employees. Since evaluation is done based on a fixed time interval, considering employee performance within the time frame will bring fairness to the output.

Moreover, employee performance appraisal process need to be conducted based on a multiple year results. This will help to see the progress of employees; moreover it helps to identify once strength and weakness. With regard to this statements, most respondents (49%) disagree that employee are evaluated based on a multi year comparison. Whereas 28% agree the point and 23% keep neutral. This shows that Zemen bank is not conducting performance appraisal process based on multiple year of performance.

In order to summarize, employees are evaluated based on comparing with their coworker. But employees are not evaluated based on multiple year of performance results and it focus only a few days of performance, we can say that there is process related problem at Zemen bank,

# 3.2.3.3. Respondent Opinion on Instrument/Form Problem

Instrument / form related problems are mainly related with the quality of the criteria to address employees' diversity, their capability to measure the employees' job duties and responsibilities specifically and weight differentiation given to each criterion.

The following Table shows that the level of instrument/form related problem of performance evaluation process at Zemen bank.

Table 7 Respondent opinion on instrument/form problem

Items		ongly Jree	Ą	gree	Neutral		Disagree		Strongly Disagree		Total	
	No.	% age	No.	% age	No.	% age	No.	% age	No.	% age	No.	% age
There is no Weight Differentiation to the factors	11	21%	7	13%	4	8%	26	49%	5	9%	53	100%
It address diversity	5	9%	7	13%	3	6%	27	51%	11	21%	53	100%
The criteria's measures the employees' job duties and responsibility specifically.	7	13%	11	21%	14	26%	9	17%	12	23%	53	100%

Analysis of the opinion revealed out that majority of the respondents (58%) disagree that the performance criteria are weighted equally, (i.e. there is weight differentiation), whereas 34% of the total respondents agree and 8% keep neutral. This implies that there is weight differentiation at Zemen bank. This is a positive parts of the bank performance evaluation criteria, since weight should be given based on their relative importance in a particular job.

A significant amount of respondents (72%) disagree that the criteria address employee diversity. Insignificant amount of respondents agree the statement. Criteria should be directly related with employee job duties and responsibility. Zemen bank evaluation criteria forms should be designed based on employees' job duties and responsibility. Since employees are doing different job, the evaluation criteria need to address employees' diversity.

In addition the criteria do not measure the employees' job duties and responsibility. They are not specific ad directly related with the job. According to the respondents, 40% of the total respondents agree the above statement. 34% of the total respondents disagree and 26% are neutral with the item. Therefore, Zemen bank performance evaluation criteria do not address employees' job duties and responsibilities specifically.

In general, there is weight differentiation on performance evaluation criteria; this is a positive side of the evaluation criteria of the bank. But those criteria's do not address diversity of the employees. Since all employees are not doing the same job and they are not in the same position; performance evaluation criteria's need to addresses such diversities. Moreover the criteria's are not specific and directly related to the job.

## 3.2.3.4. Employees' Opinion on Employees Related Problem

Some of the factors categorized under employees' related problems of performance evaluation are; decrease of employees' morale due to unfair performance evaluation practices, retention of employees due to unfair performance result and unwillingness of managers to hear from their employees (i.e. Existence of one way communication only).

Below is a summary of employees' related problem of performance evaluation process at Zemen bank.

Table 8 Respondents opinion on Employees / Subject Related problem

ltems .	Strongly Agree		Agree		Neutral		Disagree		Strongly Disagree		Total	
	No.	% age	No.	% age	No.	% age	No.	% age	No.	% age	No.	% age
Employees are allowed to give feedback	7	13%	12	23%	11	21%	17	32%	6	11%	53	100%
Employees Morale has been decreases.	10	19%	16	30%	14	26%	7	13%	6	11%	53	100%
Employees are leaving the bank.	10	19%	14	26%	17	32%	7	13%	5	9%	53	100%

As can be seen from the above Table, 11% of the total respondents strongly disagree that employees do have a chance to give feedback at the time of evaluation (i.e. there is only one way communication). While 32% disagree the statement. On the other hand, 13% strongly agree and 23% agree the statement. But 21% are neutral to the item. In order to put two way communications in to practice, managers are responsible to give time to hear employees' feedback about evaluation process. From the above Table, it is clear that number of percent of respondents oppose the existence of two way communication.

In addition, 49% of the respondents agree that employees' morale has been decreased due to unfair performance evaluation practice by the bank; moreover 45% of the total respondents agree that employees are leaving the bank due to unfair performance evaluation result. Whereas 24% of the respondents disagree that employees morale degrease due to unfair performance evaluation result and 22% of them agree that employees are not leaving the bank due to unfair result. Insignificant amount of respondents are neutral to the statements. This implies that employee retention rate due to unfair performance result at Zemen bank is high. As a result of this, Zemen bank losses potential employees and second recruitment and training cost will be high.

In general, employees are not given enough time to give feedback at the time of conducting performance appraisal, moreover employees' morale has been decrease due to unfair performance evaluation result and they are leaving the bank.

# 3.2.3.5. Employees Opinion on Managers/Execution Problem

Managers related problems of performance appraisal are the most serious parts of performance appraisal processes. Most of the time employees are blaming their managers for their poor performance results. Practically there are many dimensions that managers can be a cause for ineffective performance appraisal practices. Some of them are, Lack of managers training, managers do not know whom they evaluate, being managers the only person in the evaluation process and improper documentation of employees' performance record are some of the problem related with managers.

The following Table shows the degree of managers related problem existence at Zemen bank (see Table 3.6. below).

Table 9 Respondents opinion on Manager/Execution Related problem

	Criteria's		ongly pree	Agree		Ne	Neutral		agree	Strongly Disagree		Total	
	Official 3	No.	% age	No.	% age	No.	% age	No.	% age	No.	% age	No.	% age
9	lanager avoid giving result which has a egative effect	6	11%	12	23%	22	42%	10	19%	3	6%	53	100%
	lanager gives ual result to all	1	2%	13	25%	13	25%	19	36%	7	13%	53	100%
	Different aluation result mong different dept	22	42%	16	30%	7	13%	5	9%	3	6%	53	100%
	manager evaluate mployee who do not know them	7	13%	15	28%	15	28%	10	19%	6	11%	53	100%
th	upervisors are ne only person the evaluation	7	13%	17	32%	12	23%	13	25%	4	8%	53	100%
Ма	anager are well trained	4	8%	12	23%	14	26%	13	25%	10	18%	53	100%
	lanagers keep file on nployees work to evaluate	3	6%	11	21%	12	23%	17	32%	10	19%	53	100%

As the Table indicates above, most respondents agree that supervisors are the only persons in the evaluation process. Almost 13% of the respondents' strongly agree and 32% agree this statement. Whereas a total of 33% disagree the point and 23%

are neutral to the item. In real world it is often unrealistic to presume that one person can fully observe and evaluate employees' performance. Therefore, Zemen bank needs to improve the practice of one person performance appraisal strategy. If this strategy continuous, Zemen bank performance appraisal strategy will be high attacked by subjectivity and bias.

On the basis of managers training, also 43% of the total respondents disagreed that managers are well trained to evaluate employees' performance. Whereas 31% agree the issue and 26% are neutral to the statement. This indicates that managers training and development program based on performance assessment of employees is not satisfactory. If managers/evaluators are not well trained, they will face difficulties on how to enhance the career path of their employees and potential employees start to look for other opportunities in the market.

Analysis of the opinion of the respondents revealed out that managers do not have trended of filing employees' performance record. Large amount of respondents (51%) agree this point. Relatively insignificant amount of respondents oppose this issue. 23% of the respondents are neutral to the item. From this, we can conclude that managers are not filing employees' performance record properly. This shows that managers are evaluating employees based on events that occur during last few months rather that the entire year.

Moreover, the study shows that there is significant deviation among departments in evaluating employees. Respondents agree that there is different evaluation result among departments. 72% of the respondents agree this statement. 15% only disagree and 13 keep silent. This indicates that there is inconsistency across managers. The evaluation result of employees is highly dependent on managers' behavior.

On the contrary, respondents agree that managers avoid giving result which has a negative effect on employees. 34 of the respondents agree this statement. While 25 disagree and 42% keep silent. Since one of the main objectives of performance evaluation is to encourage employees, managers are protecting their employees from being discourage to some extent. This can be seen as a good part on managers.

In order to summarize, there is high rate of manager's related problem at Zemen bank. In view of the fact that, managers are the only evaluator, Managers skill of evaluating employees is not supported by training or development program, managers do not keep employees performance record, moreover existence of inconsistency across managers are core reasons to managers related problem.

# 3.3. Respondents/Managers Opinion towards the Core Reason Performance Appraisal is Conducted at Zemen Bank

Managers have agree that the core reasons to conduct performance evaluation at Zemen Bank are; to reward and promote employees; to encourage employees strong side and to improve their weakness; to create positive competitive environment among employees; for bonus and salary increments; to give feedback about employees performance; to improve work efficiency; to identify employees training and development needs; and to increase overall productivity of the organization.

According to the information gathered from the respondents/employees, the reasons that performance evaluation conducted at Zemen Bank are directly in line with thoughts of different authors explained in the literature parts of this research. Therefore Zemen Bank is practicing performance evaluation processes reason in accordance with the rule and regulation of the bank and general guide lines. Moreover, both managers and employees respond towards the core reason of conducting PA are parallel.

# 3.4. Respondents/Managers Opinion about the Real Problem Zemen Bank is facing in Relation with Performance Appraisal.

According to the respondents/employees the most serious problems Zemen Bank is facing with performance appraisal are, subjectivity, recency error, senior gives priority in the evaluation result, the criteria's are not related with the employees job duties and responsibility, the existence of quota system, lack of evaluators training, highly influenced by friendship, few mistakes affect employees performance result, lack of two way communication, inconsistency among managers, some managers do not know whom they evaluate, discrimination among employees, and evaluators are

not accountable for their wrong performance evaluation practice are some of highly mention problems exists at Zemen Bank. Whereas insignificant number of respondents among managers group on the contrary says that there is no problem related with performance evaluation practice. Performance evaluation practice problems are inherent in every organization that it will be beyond the reality to say that there is no PA related problems.

To summarize, Zemen Bank is facing very serious problems related with performance appraisal. Almost all problems explained in the literature review part have existed in Zemen bank. If these problems persist, Zemen bank is in a position to loss the main objectives of performance appraisal, employees' morale will decrease due to unfair performance evaluation practices done by the bank, employees' retention will increase, due to this Zemen bank training and development costs will increase at significant amount, moreover Zemen Bank will loss industries competitive power since employees are the core reasons to gain competitive advantage in the banking industry.

# 3.5. Respondents/Managers Suggestion about Performance Appraisal Practices at Zemen Bank

Some of the solutions provided by the respondents about performance appraisal practice of Zemen Banks are, the performance evaluation practice should consider employees whole year performance rather than a specific day performance, employees morale should be given priority at the time of giving feedback and conducting performance evaluation result, there should be mechanism that employees also evaluate their managers, two way communication need to be implemented, there should be independent part whom revise the evaluation result given by the evaluator in order to avoid subjectivity and bias, the criteria need to be in line with the employees job duties and responsibility, the quota system should be avoided and the evaluator should take training.

The comments give by respondents are goods and directly related with the problem exists at Zemen Bank; Moreover they are simple and can be implemented by the bank. Zemen Bank is required to do so in order to avoid the above mentioned problems at early stage.

#### CHAPTER FOUR

### **SUMMARY, CONCLUSIONS AND RECOMMENDATIONS**

Based on the results of the study obtained through the questionnaire distributed to 53 employees of Zemen Bank and the interview conducted with the human resource manager and some evaluators, the following Major findings, Conclusions and recommendations are made.

# 4.1. Summary of Major Findings

Below are summary of findings gathered from the respondents about performance appraisal at Zemen bank.

According to the HR Manual and information gathered from respondents, the MAIN reasons that performance appraisal is conducted at Zemen Bank are, for bonus and salary increments, to reward and promote employees, to encourage employees strong side and to improve their weakness, to create positive competitive environment among employees, , to give feedback about employees performance, to improve work efficiency, to identify employees training and development needs, and to increase overall productivity of the organization.

According to the information gathered from the respondents', performance evaluation criteria are in line with the bank goal and objective and point given to each criteria are in accordance with the general rule of performance evaluation criteria. But, these criteria are not specific and directly related with the employees' job and duties, they are not sufficient enough to measure the performance of an individual.

Employees are not aware of how the performance appraisal process is conducted. Managers do not give time to hear and address employees' problem. There is no practice of revising the performance appraisal by independent committee. Self assessment is not allowed. Moreover, the frequency of feedback given during a job is not consistent. In addition, feedback is given mostly at the time of performance appraisal only.

In relation with most serious performance appraisal problems, respondents' opinion confirms that managers are not accountable and employees are not getting feedback frequently.

There is a good sign that employees are evaluated by comparing with their coworkers. But respondents have assured that performance appraisal do not focus on the whole year performance as well as performance appraisal is not conducted based on multiple year result.

One of the positive sides in relation with instrument/form related problem of the bank is that weight differentiation is given to each criteria. But the criteria's do not address employees' diversity.

As respondents asserted through interview and secondary data review (employees' record), employees' morale has decrease and employees are forced to leave the bank due to unfair performance appraisal practices. Moreover, employees are not allowed to give feedback during performance appraisal.

In order to summarize, there is high rate of manager's related problem at Zemen bank. In view of the fact that, managers are the only evaluator, Managers skill of evaluating employees is not supported by training or development program, managers do not keep employees performance record, moreover existence of inconsistency across managers are core reasons to managers related problem.

### 4.2. Conclusions

On the basis of major findings the following conclusions are forwarded.

As asserted by Longenecker et al (1999), bonus and salary increments are under the main purpose of performance appraisal of the bank. The next highly rated purposes of performance appraisal are to promote and to give recognition to employee. In general, almost all mentioned purposes of performance appraisal in the literature parts of the research are noticed by respondents.

Since the performance evaluation criteria are not specific and directly related with the employees' job duties and responsibility, it is difficult to say that employees are evaluated properly. The performance evaluation criteria are not capable enough to measure employees output since the evaluation forms are the same to all ratees.

As per the respondents' opinion, we can conclude that the performance appraisal process is not in accordance with the general guide lines stated in the literature parts. As asserted by Islam et al (2005), we can say that the performance appraisal process is subject to bias, subjectivity and lack rigors. Moreover, employees are thinking that their current level of performance is acceptable in the organization and they are not putting extra effort to improve, since the frequency of feedback is poor.

Since mangers are not accountable to their wrong feedback, employees' morale will decrease and forced to leave the organization. Moreover the overall performances of the organization will decreases. In addition, if feedback is not given at the right time, the bank losses the benefit to get the most out employees.

Performance appraisal process is conducted based on events that occur during a few period of time and each year performance appraisal process stands alone. Therefore, we can conclude that employees' performance improvement is not in to consideration while evaluating employees.

Since employees' job duties and responsibilities are different according to their profession, it is good that weight given to each criteria is in accordance with their relative importance to a particular job. Whereas, the same appraisal forms are applied to large but not homogeneous group of employees, the assessment form do not fit to measure employees jobs.

According to the information gathered from employees through interview and secondary data review (Employees record), Zemen bank is losing industries competitive power, Since employees' morale has decreased and employees are leaving the bank due to unfair performance appraisal practice. Moreover, employees' involvement in performance appraisal is not at the required level, hence employees are not allowed to give feedback during performance appraisal.

Further, Zemen bank is highly attacked with managers' related problem. The evaluation processes is subject to bias and subjectivity, since performance appraisal

is done by mangers only. Employees' enhancement towards career path is less because managers are not well trained. Inconsistencies among mangers reflect that there is wrong competition among departments. Moreover, two way communications is not implemented well; hence managers do not know their employees.

#### 4.3. Recommendations

On the basis of the findings and conclusions reached, the following recommendations are forwarded in order to improve the performance evaluation practice of Zemen Bank.

- 1. In order to improve the performance evaluation criteria, the human resource department needs to design specific and directly related criteria that can measure employees' job duties and responsibility specifically.
- 2. The human resource department needs to create awareness among employees about how the performance evaluation process is conducted. Time need to be given to hear and address employees related problems. The departments should implement mechanism (Such as, Suggestion Box, Employees interview, Developing and distributing questionnaires' etc) to address employees to hear performance related issues. Moreover the existence of quota system and giving priority to senior in the performance result to award/promote employee should be avoided hence such system create bias among evaluators to evaluate properly and will be difficult to differentiate good performer with weak performer.
- 3. There should be a way that allows employees to evaluate themselves; thus the bank should introduce self evaluation. This could be done by giving a chance to employees to fill the evaluation form with their arguments prior to conducting the final result. These will help the bank to consider forgotten performance of employees.
- 4. The department needs to implement a system that will help to evaluate the performance result prior to conducting the final result to employees. This will

- help to avoid subjectivity and any bias. Moreover managers will be accountable for their performance evaluation practices.
- 5. The performance evaluation system of the bank should be designed in a way that feedback is given during the course of performing a job; the back should avoid giving feedback at the time of conducting performance evaluation only. Feedback should be continuous. This will improve employees' performance as well as help them to know their strength and weakness. Moreover, feedback given at the time of conducting the performance appraisal needs to be constructive to employees and two ways communication system should be implemented.
- 6. In order to minimize managers related problems; managers need to be well trained in evaluating and giving feedback, Managers do not have to evaluate alone. There should be independent department that control evaluators for their evaluation practice.
- 7. Managers should keep file on employees' performance record, and there should be "benchmark" numbers to avoid managers' inconsistency. By doing this, the bank can easily avoid managers related problem of performance evaluation process.

### Reference

- ✓ Anthony, Perrewe and Kacmar, (1999). *Human resource management: A strategic approach*. New York: Harcourt Brace.
- ✓ Armstrong, M. (2003). A handbook of human resource management practice.

  London: Kogan Page.
- ✓ Aswathappa, A. (2002). Human Resource and personnel management: Text and Cases, 3rd edition, Tata McGraw-Hill Publishing Company Limited, New Delhi.PP.199-231
- ✓ Bohlander, G. and Snell, S. (2004). *Managing human resources*. Mason, Ohio: South-Western.
- ✓ Campbell, D. J. and Lee, C. (1988). 'Self-appraisal in performance evaluation:

  Development versus evaluation'. Academy of Management Review, 13, 302-3 14.
- ✓ Cascio, F.W. (2003). Managing *Human Resources: Productivity, Quality of work life, profits*, 6th edition, McGraw –Hill Higher Education. PP.328-369.
- ✓ Cleveland, J. N., Murphy, K. R., & Williams, R. E. (1989). *Multiple uses of performance appraisal: Prevalence and correlates*. Journal of Applied Psychology, 74(1), 130–135.
- ✓ Clinton O. Longenecker (1997). Why managerial performance appraisals are ineffective: Causes and Lessons. Career Development international, Vol.2, 5, pp. 212-218.
- ✓ Cynthia, L. (1985). *Increasing performance appraisal effectiveness: Matching Task types, Appraisal Process, and Rater Training.* The Academy of Management Review, Vol.10, No.2, PP.322-331.
- ✓ Deborah F. Boice and Brian H. Kleiner (1997). *Designing effective performance appraisal system*. Work study, vol.46, No.6. PP.197-201
- ✓ Dessler, G. (2000). Human resource management. New Jessey: Prentice Hall.
- ✓ Dewakar G (2009). *Human Resource Management*. Delhi.PP.26-60
- ✓ Facteau, C. et al (1998). *Reactions of leaders to 360-degree feedback from subordinates and peers*. Leadership Quarterly, 9(4), 427–448.
- ✓ Francis O. Boachin Meansah (2012). "Employees perception of performance appraisal system". International Journal of Business and Management, Vol. 7 pp 73-88

- ✓ Gilbert, C. A. (1994). Performance Appraisal: A Process Analysis of Existing

  Methods and Trends [Online]
- ✓ Grote, D. (1996). *The complete guide to performance appraisal*. New York: American Management Association.
- ✓ John Edmonstone (1996). *Appraising the state of performance appraisal*. Health Manpower Management, vol.22, No. 6. pp 9-13.
- ✓ Ivancevich, J.M. (2004). *Human Resource Management*. 9th edition, McGraw-Hill/Irwin Companies, New york.pp.255-287.
- ✓ Linda S.Pettijhon (2001). Performance appraisals: usage, criteria and observations. Journal of management development, vol.20, No.9, pp.754-771.
- ✓ London, M. (2003). *Job feedback: Giving, seeking and using feedback for performance improvement.* Mahwah, NJ: Lawrence Erlbaum.
- ✓ Longenecker, C. O. (1997). Why managerial performance appraisals are ineffective: Causes and lessons. *Career Development International*, 2(5), 221–218.
- ✓ Longenecker, C.O. and Fink, L.S. (1999) "Creative Effective Performance Appraisals," Industrial Management, 18-23.
- ✓ M.S. Susan Taylor, et.al. (1995). Due process in Performance appraisal: A Quasi Experiment in procedural justice, Administrative science quarterly, Vol.40, No.3. (sept., 1995), pp.495-523
- ✓ Mark Cook (1995). Performance appraisal and true performance. Journal of managerial philosophy, Vol.10, No.7, pp.3-7
- ✓ Michael Beer (1987). "Performance appraisal", In Lorch, J.(Ed). Hand book of organizational Behavior, Prentice Hall, Englewood, Cliffs, NJ, pp.286-299
- ✓ Michael K.Mount (1984). Satisfaction with the performance appraisal system and Appraisal discussion. Journal of occupational Behavior, Vol.5, No.4. (oct., 1984), pp.271 279.
- ✓ Milkovich, G. M., & Boudreau, J. W. (1997). *Human resource management*. Chicago: Irwin.
- ✓ Murphy, K. R., & Cleveland, J. N. (1995). *Understanding performance appraisal:*Social, organizational and goal-based perspectives. California: Sage,
  Thousand Oaks
- ✓ Rafikul Islam and Shuib Bin Mohd Rasad (2005). *Employee Performance Evaluation By Ahp:* A Case Study, Honolulu, Hawaii, July 8-10, 2005

- ✓ Roberts, G. E., & Reed, T. (1996). *Performance appraisal participation, goal-setting and feedback: The influence of supervisory style*. Review of Public Personnel Administration, 16(4), 29–60.
- ✓ Rusli Ahmad and Nur Azman Ali(2004). Performance appraisal decision in Malaysian public service, The international journal of public sector management, Vol.17, No.1, pp.48-64Saiyadain, (2003). Human Resource Management. Delhi: Tata MCGraw-Hill Publishing Company Limited.
- ✓ Schuler, R. S. (1995). *Managing human resource*. Min St. Paul: West Publishing.
- ✓ Weise, D.S. and Buckley, M.R. (1998). The evaluation of the performance appraisal process. Journal of Management History, Vol.4, No.3, PP.233-249.
- ✓ Y.Fried et al. (1999). *Inflation of subordinates' performance ratings: Main and interactive effects of Rater Negative Affectivity, Documentation of Work Behavior, and Appraisal visibility.* Journal of organizational Behavior, Vol.20, No.4. (Jul. 1999), pp.431-444.
- ✓ Zemen Bank Human Resource Manual.

#### Annex I

# St. Mary's University College School Of Graduate Studies MBA Program

### To be filled by Employees

Dear respondents, this questionnaire is designed to collect information about employees' performance evaluation practice and challenges at Zemen Bank. The information shall be used as a primary data in my research which I am conducting as a partial requirement of my study at St. Mary's university college school of graduate studies for completing my MBA.

Therefore, your genuine, honest, and prompt response is a valuable input for the quality and successful completion of the research.

#### General Instructions

- ✓ There is no need of writing your name.
- $\checkmark$  In all cases where answer options are available please tick ( $\checkmark$ ) in the appropriate box.
- ✓ For questions that demand your opinion, please try to honestly describe as per the questions on the space provided

Thank you, for your cooperation and timely response in advance

Mulugeta Hailu Workneh

1. Number of years you have worked for this organization	n (in years	s)		
2. Number of years working on this job (in years):				
☐ 4 years or more☐ 3-4 ☐ 2-3 ☐ 1-2 ☐	<1			
3. Age (in years): $\square$ 55 and above $\square$ 45-54 $\square$ 3	35-44 □	25-34		under
25				
4. Sex:□ Female □ Male				
5. Educational Qualification:				
☐ High school graduate				
☐ Technical school graduate				
☐ College Diploma				
☐ BA/BSc Degree				
☐ Masters Degree				
☐ PhD ☐ other (please state		)		
6. Have you been evaluated for the last one or two year?				
□ Yes □ No				
7. If your answer to question number 7 is yes, please con	nplete pa	ırt II		
PART II: Questions related to the practices of perform	nance Ev	aluation		
Listed below are statements about the practices a	and cha	llenges of	Emp	loyee
performance evaluation in your organization. Please inc	dicate yo	our level of	agree	ement
with the statements so that your answers to these ques	stions wil	l enable m	e to a	ssess
what you think about the practices and challenges of p	performa	nce evalua	tion in	your
organization				
	gly e	e er nor 'ee	ee	gly ree
	Strongly Agree	Agree Neither Agree nor Disagree	Disagree	Strongly Disagree
The suiterie was ideal under newforms and available or	σ '	, Z & Q		ωÖ
The criteria provided under performance evaluation are aligned with the objective and goal of the bank.				
Feedback is given to employees about their performance				
evaluation so that they know where they are.				

**PART I: Participant Information** 

1.

2

Employees are aware of how performance appraisal processes is to be conducted. Employees are well informed with performance appraisal criteria.  The standards/criteria's under performance evaluation help to measures the employees' job duties and responsibility. (I.e. they are specific and directly related to the job.)  Information provided through performance evaluation process helps to strengthen the relationship between superiors and subordinates	
with performance appraisal criteria.  The standards/criteria's under performance evaluation help to  measures the employees' job duties and responsibility. (I.e. they are specific and directly related to the job.)  Information provided through performance evaluation process helps to strengthen the relationship between superiors and	
The standards/criteria's under performance evaluation help to  4 measures the employees' job duties and responsibility. (I.e. they are specific and directly related to the job.)  Information provided through performance evaluation process helps to strengthen the relationship between superiors and	
4 measures the employees' job duties and responsibility. (I.e. they are specific and directly related to the job.)  Information provided through performance evaluation process helps to strengthen the relationship between superiors and	
they are specific and directly related to the job.)  Information provided through performance evaluation process  helps to strengthen the relationship between superiors and	
Information provided through performance evaluation process  believed through the performance evaluation process  believed through the performance evaluation process and the performance evaluation pro	
5 helps to strengthen the relationship between superiors and	
subordinates	
1 1 1 1 1	
6 While conducting the evaluation result there is enough time to	
interview to hear and address employees' problems/needs.	
Employees do have chance to evaluate themselves using	
7 proper formats. i.e. Self assessment is allowed prior to	
conduction the final result.	
Employees are given positive feedback when they do good job	
8 and constructive feedback which tell them to improve for poor	
performance during a job.	
The performance results are revised by the responsible	
9 department to avoid any bias or subjectivity.	
Employees are getting further career development or	
10 promotion as a result of their good performance. I.e.	
Performance evaluation is linked with n=benefit package.	
Employees let to know their strength and weakness during	
11 performance appraisal process.	
Employees are given feedback/comment at the time of	
conducting performance appraisal result only.	
Employees are evaluated based on their supervisor	
subjectivities. Supervisor/manger is influenced by his/her	
personal liking and disliking when evaluating employees	
performance.	

		Strongly Agree	Agree	Neither Agree nor Disagree	Disagree	Strongly Disagree
14	Supervisors are not accountable/ responsible for their wrong feedback/comments.					
15	The performance evaluation process used as punishment for poor performance.					
16	Employees' performance appraisal process is conducted based on comparing multiple years result. I.e. while evaluating employees past year's performance are used as reference.					
17	Supervisors are the only persons involved in the performance appraisal process.					
18	Employees are evaluated by comparing with their co workers.					
19	All employees are evaluated with the same forms/criteria					
20	Supervisors/managers are well trained in assessing and giving feedback.					
21	All criteria's are equally weighted.					
22	My supervisor usually keeps a file on what I have done during the appraisal period to evaluate my performance					
23	Employees are evaluated based on the last few months' performance prior to conducting the performance appraisal instead of entire year performance.					
24	My Supervisor avoids giving performance ratings which may have negative consequences for his/her subordinates.					
25	My supervisor gives equivalent performance ratings to all my colleagues in order to avoid resentment and rivalries among us					
26	There is difference in evaluation result among different department manger due to managers' behavior.					
27	Employees are evaluated with their managers who do not know their employees very well.					
28	Employees do not have the chance to give feedback while conducting the result. There is one way communication only.					

		Strongly Agree	Agree	Neither Agree nor Disagree	Disagree	Strongly Disagree
29	There is formal and informal communication about the evaluation process between supervisors and employees.					
30	Employees' are leaving the bank due to unfair performance result.					
31	Employees' morale has been decreased due to unfair performance result.					

Part III. Additional Questions
1. In your opinion, what are the core reasons to conduct performance evaluation?
3. In your opinion, what are the real problems that you observe regarding performance evolutions practices of your organization?
4. Would you please suggest if there is anything to be changed with regard to the current
performance evaluation system being used in your organization?

Thank you again for completing the questionnaire!

### Annex II

# St. Mary's University College School Of Graduate Studies MBA Program

### To be answered by HRD manager and Department Managers /Evaluators

This interview is designed to collect information about employees' performance evaluation practice and challenges at Zemen Bank. The information shall be used as a primary data in my research which I am conducting as a partial requirement of my study at St. Mary's university college school of graduate studies for completing my MBA.

Therefore, your genuine, honest, and prompt response is a valuable input for the quality and successful completion of the project.

### List of interview questions:

- 1. How is performance appraisal conducted at Bank?
- 2. What are the reasons/Purposes to conduct performance evaluation at Zemen Bank?
- 3. What are the major problems and there consequences that your department is facing with respect to performance evaluation?
- 4. What measurements are taken in order to overcome the problems?
- 5. How are employees involved in the performance appraisal processes?
- 6. Finally, is there anything that you want to comment about the performance appraisal system of your organization?

#### Mulugeta Hailu Workneh